1	
2	Climate change and African trypanosomiasis vector populations in Zimbabwe's
3	Zambezi Valley: a mathematical modeling study
4	Jennifer S. Lord* ¹ , John W. Hargrove ² , Stephen J. Torr ¹ and Glyn A. Vale ^{2,3}
5	
6	¹ Department of Vector Biology, Liverpool School of Tropical Medicine, Liverpool, United
7	Kingdom
8	² SACEMA, University of Stellenbosch, Stellenbosch, South Africa
9	³ Natural Resources Institute, University of Greenwich, Chatham, United Kingdom
10	
11	* jennifer.suzanne.lord@gmail.com

12 Abstract

13 Background

Quantifying the effects of climate change on the entomological and epidemiological components of vector-borne diseases is an essential part of climate change research but evidence for such effects remains scant, and predictions rely largely on extrapolation of statistical correlations. We aimed to develop a mechanistic model to test whether the recent increase in temperature in the Mana Pools National Park of the Zambezi Valley of Zimbabwe could account for the simultaneous decline of tsetse flies, the vectors of human and animal trypanosomiasis.

21 Methods and findings

The model we developed incorporates the effects of temperature on mortality, larviposition
and emergence rates, and is fitted to a 27-year time series of tsetse caught from cattle. These
catches declined from ~50 flies per animal per afternoon in 1990 to ~0.1 in 2017. Since 1975,
mean daily temperatures have risen by ~0.9°C and temperatures in the hottest month of
November by ~2°C. Although our model provided a good fit to the data, it cannot predict if or
when extinction will occur.

28 Conclusions

The model suggests that the increase in temperature may explain the observed collapse in tsetse abundance and provides a first step in linking temperature to trypanosomiasis. If the effect at Mana Pools extends across the whole of the Zambezi Valley then transmission of trypanosomes is likely to have been greatly reduced in this warm low-lying region. Conversely, rising temperatures may have made some higher, cooler, parts of Zimbabwe more suitable for tsetse and led to the emergence of new disease foci.

36 Author summary

37 Why was this study done?

38	• Tsetse flies are blood-feeding insects that transmit pathogens causing fatal diseases of
39	humans and livestock across sub-Saharan Africa.
40	• The birth and death rates of tsetse are influenced by environmental conditions,
41	particularly temperature.
42	• Since 1975, mean daily temperatures at Rekomitjie, a research station in the Zambezi
43	Valley of Zimbabwe, have risen by $\sim 0.9^{\circ}$ C and temperatures in the hottest month of
44	November by $\sim 2^{\circ}$ C. These increases in temperature may have impacted tsetse
45	populations and the diseases they transmit.
46	
47	What did the researchers do and find?
48	• Since the 1960s, wild tsetse have been caught regularly from cattle for insecticide
49	tests conducted at Rekomitjie.
50	• Catches from single cattle have declined from >50 flies per afternoon prior to 1990 to
51	~0.1 in 2017.
52	• A mathematical model of tsetse population change, which included temperature-
53	dependent rates for births and deaths, suggests that the decline in tsetse is related to
54	rising temperatures.
55	What do these findings mean?
56	• Our findings provide a first step in linking the effects of increasing temperatures to
57	the distribution of diseases caused by tsetse.

If the effect extends across the Zambezi Valley then tsetse-borne disease is likely to
 have been reduced across the region. Conversely, rising temperatures may have made
 some higher, cooler areas more suitable leading to the emergence of new disease foci.

62 **Introduction**

Tsetse flies (Glossina spp.) transmit protozoa of the genus Trypanosoma that cause sleeping 63 sickness (human African trypanosomiasis, HAT) in humans. The initial phase of HAT is 64 characterised by intermittent fever and joint pains, thereafter there are sleeping difficulties 65 66 and confusion. Without treatment the disease is fatal. Parasites of this genus also cause nagana (animal African trypanosomiasis, AAT) in livestock. 67 68 69 In 2015, HAT was responsible for ~202,000 DALYs (Disability-Adjusted Life Year) [1]. The 70 most recent global estimates indicate that AAT kills ~1 million cattle/year [2], with ~55 million cattle at risk [3]. 71 72 In addition to the DALYs resulting from HAT, AAT also has substantial impacts on human 73 health by reducing the supply of meat and milk, as well as animal draft power for crop 74 production. These losses affect not only human nutrition but also the agricultural incomes 75 76 that allow access to education and health care [4]. A study in 1999 indicated that the annual 77 economic losses from meat and milk production alone were ~US\$1 billion at current prices [5]. 78

79

In Africa, there has been an increase in temperature of *c*. 1.5°C between 1900 and the 1990s [6]. However, the effects of recent and future climate changes on the distribution of tsetse and other vectors, and their associated diseases, remain poorly understood [7,8]. There is a disagreement, for example, about whether the resurgence of malaria in the East African highlands in the 1990s was caused by rising temperatures, or by increasing levels of drug resistance and decreasing control efforts [9–13]. Resolution of the debate is made more complex by the apparent absence of data on changes in vector population levels and bitingrates.

88

Increases in global temperatures since the late 1800s [14] have led to shifts in the ranges of
many animals [15]. Insects in particular are sensitive to changes in temperature, with
consequences for the transmission of vector-borne pathogens [7,16,17]. Mechanistic models,
capable of explaining how recent climate change [14] has affected vector distribution and
abundance, could be used to predict future disease risks [16], but existing studies often rely
instead on statistical correlations [8,18–20].

95

In general, the ways in which climate change will affect infectious disease burden in subSaharan Africa is poorly understood because of a lack of empirical evidence [21]. It has been
suggested that requirements for accepting a 'causal' relationship between climate change and
changes in human health outcomes for vector-borne diseases should, as a minimum, include:
evidence of biological sensitivity to climate; ii) meteorological evidence of climate change;
iii) evidence of entomological and/or epidemiological change in association with climate
change [8].

103

For vector-borne diseases, the difficulty is the ability to separate climatic effects from those originating from other environmental, ecological and sociological changes influencing the population dynamics of parasites and vectors. Contributing to this difficulty is a lack of contiguous data on vector abundance and detailed records of local climate. Work on tsetse and trypanosomiasis carried out at Rekomitjie Research Station in the Mana Pools National Park, Zimbabwe over the last 57 years provides a valuable exception to this rule, producing long-term datasets for both vector abundance and climate profile.

Importantly, the study site is located >10km inside a protected area (S1 Fig). According to 112 the World Database on Protected Areas (https://protectedplanet.net/), the Mana Pools 113 National Park, together with its adjoining Sapi and Chewore Safari Areas and the adjacent 114 Hurungwe Safari Area, has a total area of 9660km². It has been free of agricultural settlement 115 since 1958 when the people living there were relocated [22]. Since then the combined area 116 117 has been protected against settlement, agriculture, illegal hunting and logging and was designated a UNESCO World Heritage Site in 1984. In this area, HAT occurs and tsetse 118 119 populations have not been exposed to any form of control. In addition, being situated in a protected area, the region has not been subject to other deliberate environmental or 120 sociological change. Analyses by Hansen et al. [23] show that this area experienced <0.5% 121 woodland loss between 2000 and 2010, with the majority of the 30m x 30m pixels in the 122 Hansen et al. dataset within Mana Pools consisting of at least 10% wooded cover (S1 Fig). In 123 addition, an aerial survey for elephant and buffalo in 2014 [24] indicated that in the 200km² 124 around Rekomitije there was an average of 1.6 elephants and 7.3 buffalo per km². Vale *et al.* 125 [25] showed that \sim 2 elephants per km² can provide about half of the diet of savanna species 126 of tsetse and can support a population of flies even when alternative hosts are heavily 127 depleted. 128

129

The data available therefore provide the possibility of developing temperature-driven models for tsetse population dynamics. Such models could be used to predict the present and future distribution of tsetse in Africa. Given that there is never any cyclically transmitted African trypanosomiasis without the presence of tsetse, such models will provide a more powerful approach for estimating changes in the distribution of human and animal trypanosomiases.

Tsetse flies are poikilotherms and their development and mortality rates are dependent on 136 temperature [26–30]. We aimed to use data on temperature and tsetse abundance at 137 138 Rekomitjie to test whether observed increases in temperature over recent years are sufficient to explain the observed decline in the local tsetse population since the 1990s. To do this we 139 used a mechanistic model of tsetse population dynamics that incorporates the effect of 140 temperature on adult and pupal mortality and rates of larval deposition and pupal 141 142 development, established from laboratory and field studies [26-30]. We fitted the model to a 27-year dataset of *Glossina pallidipes* numbers caught from bait oxen. 143

144

145 Methods

The methods for the production of data for tsetse and climate were not guided by an analysis plan for the present study. Instead, the climate data were produced as a standard procedure at the research station over the past 59 years, and the tsetse data were obtained from previous studies [31–36].

150

151 Temperature data and analysis

Daily records of rainfall and minimum and maximum temperature have been kept at 152 Rekomitjie since 1959. Staff at Rekomitjie, operating in accord with directions from The 153 Zimbabwe Department of Meteorological Services, made recordings at 0700h each day from 154 maximum and minimum mercury thermometers housed in a Stevenson screen. The location 155 156 of the screen is at 16° 10'S, 29° 25'E, altitude 503m. To quantify changes in the mean temperature over time, we first calculated mean monthly temperatures between October 1959 157 and June 2017. Then, using a reference period between January 1960 and December 1989, 158 159 we calculated monthly temperature anomalies by subtracting the reference mean from the 160 actual mean. We smoothed the temperature anomaly data using a five-year running mean, as

done for previous analyses of regional and global changes in temperature [37–39]. In
addition, a time series linear regression model was fitted to the mean monthly temperature
data using the tslm function from the forecast package [40] - a wrapper for fitting linear
models allowing for a trend variable. We subsequently employed the fitted trend to estimate
the change in monthly temperature, from the peak in 1975 to the peak in 2017, and the 95%
prediction intervals, using the forecast function in R [41].

167

168 Tsetse data

169 Sampling of tsetse at Rekomitjie, in pursuit of various ecological and behavioural studies, has suggested a decline in tsetse abundance in the last two decades. It is difficult to interpret the 170 catches confidently since they have been made by widely different methods at irregular 171 intervals. From 1966, however, fed female G. pallidipes have been collected from stationary 172 oxen at Rekomitjie, with the sole original aim of providing test insects for bioassays [31–36]. 173 Since these collections were made using a single sampling system, run at approximately the 174 same time each day, the change in the numbers collected offer an indication of the extent of 175 the decline in tsetse abundance over recent decades 176

177

Catches were made for three hours in the afternoon during the period of peak tsetse activity
[42]. Each collection team comprised two hand-net catchers and an ox, operating within 2km
of the research station. Each team operated at least 200m from other teams, in areas chosen to
maximise catches in accord with seasonal changes in the distribution of tsetse between
vegetation types [43]. In the 1960s it was usual for each team to take enough tubes to collect
a maximum of about 50 flies each day. This quota was set in consideration of the minimum
expected catch at that time, and has been maintained at this level ever since, even though it

has proved impossible to meet the quota in the last two decades. Daily records are available
since 1990 for the number of catching teams employed, and for the catch of each team. The
monthly averages of the number of flies caught per team per day are taken as indices of fly
abundance. Prior to 1990, tsetse catches regularly reached the upper limit of 50 flies:
thereafter this hardly ever occurred. Fitting the model only to catch data for the period after
1990 ensured that there was no truncation of data used in the fitting procedure.

191

Modeling tsetse population dynamics

Tsetse females give birth, approximately every 7-12 days [28], to a single larva, which 193 immediately burrows into the ground and pupates, emerging 30-50 days later as a full-sized 194 195 adult [44]. Female adult flies can live up to 200 days [45]. As quantified by researchers in the laboratory and field, larviposition and pupal emergence rates are dependent on temperature, 196 as are the mortality rates of both pupae and adults [26–28,30]. Preliminary analyses 197 suggested that the inclusion of temperature-dependent mortality rates was sufficient to model 198 the observed decline. In response to suggestions from peer reviewers, we also re-analysed the 199 200 data using models which included functions for temperature-dependent laviposition and pupal 201 emergence rates.

202

Hargrove [29,30], using data from mark-recapture experiments on Antelope Island, Lake Kariba, Zimbabwe, showed that for *G. pallidipes*, female adult mortality increases with temperatures above 25°C. In our ordinary differential equation (ODE) model of tsetse population dynamics, described below, we therefore model female adult losses per day (μ_A) due to temperate-dependent mortality using:

208
$$\mu_A = \begin{cases} a_1 & T \le 25\\ a_1 \exp(a_2(T - T_1)) & T > 25 \end{cases}$$
(1)

where *T* is temperature in °C; T_1 is not a parameter, but a constant set to 25 to ensure a_2 is in a convenient range.

211

For pupae, the relationship between mortality rate per day (μ_p) and temperature was quantified by Phelps [27] in the laboratory. The data from these experiments show that pupal survival to adulthood is highest for temperatures between about 20 and 30°C. As temperatures depart from this range the mortality rises sharply, resulting in a U-shaped curve.

This form of relationship has also been documented for various other insects [46], and for

tsetse, can be suitably modelled using the sum of two exponentials:

218
$$\mu_p = b_1 + b_2 \exp(-b_3(T - T_2) + b_4 \exp(b_5(T - T_3)))$$
(2)

where *T* is temperature in °C. T_2 and T_3 are not parameters, but constants chosen to ensure that the coefficients b_3 and b_5 are in a convenient range and were set to 16°C and 32°C, respectively.

222

223 Phelps also quantified the daily rate of pupal development (β) in *G. m. morsitans*, as a 224 function of constant temperature in the laboratory, fitting the data using the function [29]:

225
$$\beta = c_1 / (1 + \exp(c_2 + c_3 T))$$
(3)

where for females the fitted estimates were $c_1 = 0.05884$, $c_2 = 4.8829$ and $c_3 = -0.2159$.

227 The effects of temperature on pupal development and mortality rates in the laboratory are

supported by work showing similar effects in the field [44,47–49].

Lastly, using ovarian dissection data from marked and released *G. m. morsitans* and *G. pallidipes* at Rekomitjie, Hargrove [28] showed that the larviposition rate per day (*ρ*)
increases linearly between 20 and 30°C. We therefore assume a linear increase in
larviposition rate with temperature using the equation:

234
$$\rho = d_1 + d_2(T - T_4)$$
 (4)

where T_4 was set to 24°C. The time taken for a female tsetse to produce her first larva is longer than for subsequent larvae. Accordingly the values for d_1 and d_2 in Eq. 4 are lower for nulliparous females, ($d_1 = 0.061$ and $d_2 = 0.002$ (ρ_n)) than for parous females ($d_1 = 0.1046$ and $d_2 = 0.0052$ (ρ_p)) [30].

239

Considering the above temperature-dependent processes, and using the outlined functions for the five parameters μ_A , μ_P , β , ρ_n and ρ_p , we model changes in the numbers of *G. pallidipes* female adults (*A*) and pupae (*P*) using three ODEs:

243
$$\frac{dP}{dt} = \rho_n A_n + \rho_p A_p - (\beta + \mu_P + \delta P)P$$
(5)

$$\frac{dA_n}{dt} = \frac{\beta}{2}P - (\mu_A + \rho_n)A_n \tag{6}$$

245
$$\frac{dA_p}{dt} = \rho_n A_n - \mu_A A_p \tag{7}$$

Pupae are produced by nulliparous (*A_n*) and parous (*A_p*) adult females at rates *ρ_n* and *ρ_p*respectively. Losses from the pupal stage are due to pupae emerging as nulliparous adults
(*A_n*), at rate β/2, to density-dependent mortality, with coefficient δ, and mortality μ_P. Losses

from the nulliparous adult stage are due to first larviposition at rate ρ_n and mortality (μ_A), assumed equal for both nulliparous and parous females.

251

252 Model fitting

As initial starting estimates for the parameters in the model described in Eq. 5-7, we used the published [26,28,30] fitted values for larviposition and pupal emergence rates as described above (Eq. 3, 4). For adult and pupal mortality, we fitted the functions in Eq. 1 and 2 to published data – described above and in [27,29,30] - using nonlinear least squares regression.

It was not necessary to vary all parameters in the ODE model to get a reasonable fit to the 258 bioassay catch data. The only parameter in the population dynamic model for which we did 259 260 not have an initial starting estimate from published data was the density-dependent mortality coefficient (δ). For model fitting, therefore, we first allowed only this parameter to vary, 261 while keeping all other parameter values fixed. We then fitted the model to the average 262 monthly tsetse catches allowing just δ and the parameters for adult mortality - a_1 and a_2 to 263 vary, followed by those for just pupal mortality $(b_1 - b_5)$ and lastly for δ and both mortality 264 functions. For pupal mortality, it was only necessary to fit b_1 , b_3 and b_5 in the ODE model. 265 Model fits to the data were compared using Akaike's Information Criterion (AIC). 266

267

As a preliminary to the data fitting procedure, the model was run for five years prior to the start of the first month of available temperature data in October 1959 using the average monthly temperatures from October 1960 to September 1961 because we did not know initial starting values for numbers of pupae, relative to the numbers of fed female adults caught. The initial number of parous adults and pupae at time t = 0 was set to 100 and the number of nulliparous adults to 25 and the model solved at monthly time steps for comparison with values from the bioassay catch data. We fitted the model to the data using maximum
likelihood, assuming the data were Poisson distributed. For 80% of months, the variance to
mean ratio for the daily catch data was less than 1.5, and between 1.5 and 4.0 for 20%,
indicating that in most circumstances the variance was approximately equal to the mean. For
each set of parameters, we first estimated parameter values using the stochastic simulated
annealing algorithm [50] and then used updated parameter estimates in a final fit using
Nelder-Mead [51].

281

282 A penalty was incurred for parameter estimates of a_1 greater than 0.04 or less than 0.01 ensuring baseline adult mortality was within biologically reasonable limits, by stopping the 283 function before computing the likelihood and automatically assigning a high negative log 284 likelihood value [45]. A penalty was also incurred for model fits where on average there were 285 fewer than 50 tsetse between January 1965 and December 1984 since during that period 286 sampling teams consistently obtained their quota of 50 flies in an afternoon. Confidence 287 intervals (95%) were calculated for fitted parameters using the Fisher information matrix. 288 Peer reviews noted that these confidence intervals allow for no uncertainty in the fixed 289 parameters. To explore the importance of this, we refitted the model using the upper and 290 lower limits of the 95% confidence intervals of the fixed parameters b_1 , b_3 and b_5 of the 291 function for the temperature-dependence of pupal mortality. All analyses were done in R [41] 292 293 and are available online, with all the data required to reproduce figures, at the following website: https://github.com/jenniesuz/tsetse climate change. 294

295

296 **Results**

297 **Temperature increase at Rekomitjie**

298	Although there is considerable seasonal and inter-decadal variation in temperature (Fig 1a),
299	our analyses indicate an increase of ~ 0.9° C from the peak in 1975 to the peak in 2017 (Fig
300	1b). This increase is not even across the year, being greatest in November when temperatures
301	are already highest. During this month, mean daily temperatures have increased by $\sim 2^{\circ}C$
302	between 1975 and 2017 (Fig 2). In addition, the number of consecutive years in which the
303	hottest mean monthly temperature has been above 30°C has increased since 1990 (Fig 1a).
304	
305	
306 307 308 309 310	Fig 1. Temperature at Rekomitjie. a) Monthly mean temperatures. Horizontal line at 30° C highlights the increase in the number of consecutive years during the hot-dry seasons in which mean monthly temperatures have exceeded this level. b) Five-year running mean monthly temperature (°C) anomalies relative to $1960 - 1990$ reference period.
311	
312 313 314 315 316	Fig 2. Increases in mean daily temperature between 1975 and 2017 calculated for each month of the year. Estimated using time-series linear regression. Segments are 95% prediction intervals. All months except January and April had a statistically significant (p <0.05) increasing trend between 1975 and 2017.
317	
318	Modeling changes in the G. pallidipes population
319	Tsetse flies are poikilotherms and their development and mortality rates are dependent on
320	temperature [26–30]. We used four temperature-dependent functions, with starting
321	parameters estimated from fits to published data for pupal and adult mortality, larviposition
322	and pupal emergence rates (Fig 3, Table 1), in an ordinary differential equation model of
323	tsetse population dynamics (Eq. 5-7).
324	
325 326 327 328	Fig 3. Fitted temperature-dependent functions. a) Adult female mortality rate per day: points – published estimates from mark-recapture experiments on Antelope Island, Zimbabwe [30]; line – fitted temperature-dependent adult mortality function (Eq. 1). b) Pupal mortality rate per day: points – published estimates from laboratory experiments [27]; line –

- 329 fitted temperature-dependent pupal mortality function (Eq. 2). c). Pupal emergence rate per
- day: points published estimates from laboratory experiments; line Eq. 3 fitted as
- described in [26]. d) Larviposition rate per day: points data from published field
- experiments [28]; lines Eq. 4 fitted as described in [30]. See Table 1 for fitted parameter
- estimates of the mortality functions.
- 334
- **Table 1. Summary of model parameters.** Fixed parameter values used, and estimates (95%
- confidence intervals) from population dynamic model with lowest AIC (Fig 4). Fixed
- 337 parameters estimated from published laboratory or field data and fitted using nonlinear least-
- squares regression, fixed estimates using this method shown \pm standard error (Fig 3).

Parameter	Function or parameter definition	Estimate from fit of Eq. 1-4 to published laboratory and field data	Estimate from fit of population dynamic model
a_1	Eq. 1: adult mortality rate (μ_A)	0.027 ± 0.001	0.03365 (0.03363 - 0.03368)
a_2		0.153 ± 0.020	0.1168 (0.1166 - 0.1169)
b_1		0.0019 ± 0.0004	Fixed
b_2	Eq. 2: pupal	0.006 ± 0.001	Fixed
b_3	mortality rate	1.481 ± 0.681	Fixed
b_4	(μ_P)	0.003 ± 0.001	Fixed
<i>b</i> 5		1.211 ± 0.117	Fixed
<i>C</i> 1	Eq. 3: pupal	$0.05884 \pm 0.00289 \ (24)$	Fixed
<i>C</i> ₂	emergence rate (β)	4.8829 ± 0.0993 (24)	Fixed
С3		-0.2159 ± 0.0050 (24)	Fixed
d_I	Eq. 4: larviposition rate (ρ)	Nulliparous - 0.061 ± 0.002 Parous - 0.1046 ± 0.0004 (26)	Fixed
<i>d</i> ₂		Nulliparous - 0.002 ± 0.0009 Parous - 0.0052 ± 0.0001 (26)	Fixed
δ	Density- dependent mortality coefficient	NA	0.00002357 (0.00002349 – 0.00002364)

340 The observed decline in catches of fed female *G. pallidipes* has continued to the present day

341 and the rate of decline has accelerated since 2010 to the point that teams now sometimes fail

to catch a single fly in an afternoon (Fig 4). If these catches scale roughly with the population
density of tsetse around Rekomitjie throughout the study period, the data suggest a steady
decline in numbers over the last 27 years.

345

Fig 4. Observed (points) and modelled (line) changes in numbers of *Glossina pallidipes*females caught between 1960 and 2017. Data, on log base 2 scale, from 1990 to 2017 are
average numbers caught by hand net, per afternoon, using an ox-bait. Fitted parameters are
provided in Table 1.

350

To simulate this decline, the model was run using mean monthly temperatures between 351 October 1959 and June 2017. Model fits to the monthly tsetse catch data for 1990 to 2017 352 (Fig 4) varying δ and only the adult mortality parameters a_1 and a_2 , while keeping all other 353 parameters fixed, gave the lowest AIC of 1609 and provided a reasonable fit to the data (Fig 354 4). By comparison, varying only δ , or δ and parameters in Eq. 2, or δ and parameters in both 355 Eq. 1 and 2 produced AIC values of 2867, 1789 and 1764 respectively. Fixed and fitted 356 357 parameter estimates for each function are summarised in Table 1. From 1959 until the mid-1980s, fitted model numbers of tsetse fluctuated between about 50 and 100 and then declined 358 from ~50 in 1990 to <1 in 2017, in good agreement with observed data. In addition, the fitted 359 parameters a_1 and a_2 for adult mortality as a function of temperature (Eq. 1) were similar to 360 estimates from fits to the published mark-recapture data (S2 Fig, Table 1). The main 361 difference was a higher baseline mortality for adults and a slower increase with temperatures 362 above 25°C. When we carried out the sensitivity analysis, the upper and lower bounds for the 363 coefficients for adult mortality were $a_1 = 0.024$ and 0.030, $a_2 = 0.145$ and 0.198 (S1 Table). 364 365 Between 1959 and 2017, the pupal mortality rate was usually <0.005 in the fitted model. 366

Prior to the 1990s, mortality was higher than this in October-December in 14 months over 30

368 years. Since the 1990s, the pupal mortality rate was higher than this in 31 months during the

hot-dry season, over 27 years. The hot-dry season is also the time of year when the modelled
adult mortality was highest - >0.05 day⁻¹. Adult and pupal mortalities in the fitted model were
both above these levels in October and November more frequently in years after 1990. This is
consistent with the idea that increasing temperatures during the hot-dry season are primarily
responsible for the observed decline in numbers of tsetse at Rekomitjie since the 1990s, and
particularly since 2000. Indeed, increases in mean daily temperatures have been most
pronounced in November when temperatures are already highest (Fig 2).

376

The results of the preliminary analyses using constant values for larviposition and pupal emergence rate parameters are presented in S1 Text. The model with no temperaturedependent parameters did not provide a good fit to the data and had an AIC of 6762 compared to 2523 when adult temperature-dependent mortality was included (S1 Text).

381

382 **Discussion**

While there are statistical models relating climate change to changes in vector populations [8,18–20], mechanistic models that relate climate change to data for the population dynamics of an important vector of human and animal pathogens are much less common. Our mechanistic model, incorporating the effects of temperature on mortality, larviposition and emergence rates was sufficient to explain the observed decline in numbers of tsetse. The >99% decline in numbers reported here is comparable to the effects of successful large-scale tsetse control operations conducted in Zimbabwe.

390

Hargrove and Williams [52] found, similarly, that temperature was an indispensable factor in
modeling tsetse population growth on Antelope Island, Lake Kariba, Zimbabwe. They had
access to a wide range of measures of meteorological variables, but found that once

temperature had been included in their model, the addition of any further candidate variables
including rainfall, humidity, saturation deficit and cloud cover, did not result in any
improvement in the fit to the data. At Rekomitjie, over the whole study period we had data
only on temperature and rainfall. Nonetheless, the Antelope Island study suggests that we
were unlikely to be missing other important climatological variables.

399

Our results provide evidence that locations such as the Zambezi Valley in Zimbabwe may
soon be too hot to support populations of *G. pallidipes*. Similarly, *G. m. morsitans*populations at Rekomitjie are declining and might also be close to local extinction within the
next few decades [53].

404

There are several biologically feasible reasons to expect increased tsetse mortality at high 405 temperatures. Tsetse are poikilotherms and their metabolic rate increases with temperature: 406 adult tsetse therefore utilise their blood-meal more rapidly at elevated temperature and must 407 feed more frequently. But feeding is a high-risk activity and increased feeding rates will 408 likely result in increased mortality [54,55]. Tsetse use artificial refuge sites when ambient 409 temperatures exceed 32°C [56], thereby reducing the temperatures they experience by up to 410 6°C during the hottest times of the day [57]. This behaviour reduces their metabolic rate, but 411 also reduces their chances of feeding. Hence or otherwise female tsetse have reduced fat 412 413 levels, and produce progressively smaller pupae, as temperatures increase [58,59]. This has a knock-on effect on pupal mortality because smaller pupae can suffer very high mortality at 414 elevated temperatures [47]. 415

416

417 As temperatures increase, rates of pupal fat consumption increase linearly with temperature,418 whereas pupal duration decreases exponentially. The interplay between these rates results in

fat levels at adult emergence being highest for pupae experiencing temperatures of about 419 27°C, and progressively lower as temperatures increase above this level [26,47]. Reduced fat 420 levels at adult emergence prejudice the chances of a teneral fly finding its first meal before fat 421 reserves are exhausted and the fly starves, or suffers excess mortality as a consequence of 422 taking additional risks in attempting to feed [60]. The rate at which fat is used by teneral flies 423 increases with temperature, exacerbating the above problems for the fly. There are also direct 424 425 effects of high temperature on pupal mortality such that, when they are maintained at a constant level >32°C, no pupae emerge (Fig 3b) and all are found to have died before they 426 427 utilised all of their fat reserves [47].

428

Few studies of vector-borne disease have been able to show a clear link between climate 429 change and a change in either vector or pathogen population dynamics and subsequent 430 disease burden [8]. Studies are frequently confounded by other environmental, ecological and 431 sociological factors, or the necessary empirical data are too difficult to collect. Although we 432 acknowledge that this study presents only a first step in linking the effects of climate change 433 to changes in trypanosomiasis, it suggests that climate change is already having effects on the 434 density of disease vectors. In this respect, our study contributes vital analysis of long-term 435 (>10 years) data in a region where temperatures have increased and where as a consequence 436 the dynamics of a disease vector have also changed [8]. 437

438

If these effects extend across the Zambezi Valley then transmission of trypanosomes is likely to have been greatly reduced in this region. Conversely, rising temperatures may have made some higher, and hence cooler, parts of Zimbabwe more suitable for tsetse and led to the emergence of new disease foci. There is a pressing need to quantify the magnitude and spatial extent of these changes on tsetse and trypanosomiasis.

445

446

447

448

449

450

451

While there are no data on annual incidence of HAT from Zimbabwe to compare with the long-term data on tsetse populations, in the last 20 years cases have been reported from the vicinity of Makuti [61], a relatively high-altitude site of ~1500 m, where tsetse populations are close to their low temperature limit. We are unaware of trypanosomiasis being reported from this area prior to the 1990s. This circumstantial evidence of the emergence of HAT in cooler regions of Zimbabwe raises the possibility of the resurgence of tsetse populations, and

452 rinderpest epizootic of the late 1890s, apparently because the areas were too cold. Since tsetse

then of T. brucei infections, in parts of southern Africa where they have been absent since the

453 dispersal is thought to arise through random movement [62] such a resurgence would come

454 about where diffusion took tsetse to areas that are now climatically more suitable than they

were in the recent past. Tsetse populations could only become established if, in addition,

456 there were sufficient numbers of host animals and suitable vegetation to support tsetse.

Hwange National Park in Zimbabwe and Kruger National Park in South Africa are examplesof such areas, where suitable hosts and habitat for tsetse are abundant.

459

HAT is one of several vector-borne diseases where detecting human cases is difficult even in 460 countries with relatively strong health systems. In Uganda for instance, it is estimated that for 461 every reported case of Rhodesian HAT another 12 go undetected [63]. In remote parts of the 462 Democratic Republic of Congo (DRC), Central African Republic and South Sudan, finding 463 cases is even more difficult. In these circumstances, prospects for gathering data to detect or 464 predict the impact of climate change on HAT seem poor. Models to predict where vectors 465 are abundant, supported by xenomonitoring of tsetse populations for pathogenic 466 trypanosomes [64], seems a more likely means of assessing the impact of climate change. 467

In general, if the temperature increase seen at Rekomitjie is reflected more broadly in the
region, large areas that have hitherto been too cold for tsetse will become climatically more
favourable, and could support the flies if adequate hosts were available there [65].

472

In any region where the climate becomes more suitable for tsetse, there must however be 473 adequate vegetation cover to provide shelter for tsetse. The clearing of land for agricultural 474 475 development, which is occurring at an accelerating pace in many parts of Africa [66], will reduce the vegetation cover and the densities of wild hosts, in what has been termed the 476 477 autonomous control of tsetse [67]. Any future predictions of the effects of climate change on tsetse populations and/or trypanosomiasis, should consider these other confounding effects, 478 as has been done for malaria [68]. Gething et al. [68] demonstrated that any future predicted 479 changes in malaria due to climate would likely be magnitudes smaller than changes due to 480 control and other anthropogenic factors. 481

482

Most (>95%) cases of HAT occur in Central and West Africa where the important vectors are
subspecies of *G. palpalis* and *G. fuscipes*, which are riverine tsetse. These species have very
similar physiology to the savanna species of East and Southern Africa, including *G. pallidipes*, and hence we would expect that populations of riverine tsetse would decline if
they were exposed to the temperature increases reported in the Zambezi Valley of Zimbabwe.

488

489 Over the past decade, ~ 90% of all reported cases of Gambian HAT occurred DRC [69]. For
490 the tsetse-infested regions of DRC where HAT occurs (e.g., Provinces of Mai Ndombe,
491 Kwilu and Kasai) there are no data to suggest that climate change has had an impact on tsetse
492 and HAT. For HAT foci in West Africa however, there is some evidence that climate change
493 has had an impact. First, Courtin *et. al.* [70] describe a 100km shift southwards in the

northern limit of tsetse which they attribute to drought, rising temperatures and increased 494 human density. Regions where tsetse appear to be absent include areas where sleeping 495 sickness occurred in the 1930s. Second, Courtin et al. [71] report that across West Africa the 496 more northerly foci of HAT located in Senegal, Mali, Burkina Faso and Niger have ceased to 497 be active. The authors attribute this change to increased densities of humans, anthropogenic 498 destruction of tsetse habitat and climate change. Medical surveys conducted between 2000 499 500 and 2006 did not detect any cases of HAT north of the 1200 mm isohyet and comparison of the 1200 mm isohyet for the periods 1951-69 and 1970-89 show that it had shifted south. 501

502

For tsetse-infested areas of West Africa, Courtin *et al.* suggested that it is difficult to disentangle the effects that changes in land-cover, host populations, rainfall and temperature have on tsetse populations and sleeping sickness [70,71]. Studies are further confounded by the impact of large-scale medical interventions which have led to a decline in the annual incidence of Gambian HAT across Africa [72]. With such interpretive problems there is a need for more studies of the present sort where long-term measurements of tsetse abundance are made in wilderness areas where there is little change in land-cover and host populations.

510

511 The estimated confidence intervals for model-fitted parameters, such as those for adult 512 mortality, are underestimates, in part because they incorporate only the uncertainty resulting 513 from fitting the model with fixed values for other parameters, and do not incorporate 514 uncertainty in those fixed parameters. Another problem is that we did not have sufficient data 515 to test the predictive power of our fitted model.

Our deterministic model does however provide a good fit to available data for the change in 517 tsetse abundance since 1990. Such models are less satisfactory for assessing if and when a 518 population will actually go extinct, since they predict that populations go to zero only as time 519 goes to infinity. Ideally, therefore, future modeling should adopt stochastic approaches to 520 predictions about tsetse extinction, but these would require detailed knowledge of population 521 dynamics at very low density, such as the probability that male and female tsetse will meet in 522 523 sparse populations. Unfortunately, our current knowledge of dynamics relates only to populations that are dense enough for convenient study. Nonetheless, present modeling does 524 525 raise the possibility of the extinction of the Rekomitjie tsetse populations, particularly if temperatures increase further. Future research could also make use of the fitted model to 526 make spatially explicit predictions about tsetse population dynamics for other regions of 527 Zimbabwe and east and southern Africa under future predicted climate change scenarios. 528

529

530 Acknowledgements

The authors are grateful to Mr William Shereni, Director, Tsetse Control Division,
Zimbabwe, for his continued support of our work and for making available the resources at
Rekomitjie Research Station.

535 **References**

557

8.

536	1.	Kassebaum NJ, Arora M, Barber RM, Brown J, Carter A, Casey DC, et al. Global,
537		regional, and national disability-adjusted life-years (DALYs) for 315 diseases and
538		injuries and healthy life expectancy (HALE), 1990–2015: a systematic analysis for the
539		Global Burden of Disease Study 2015. Lancet. 2016;388: 1603–1658.
540		doi:10.1016/S0140-6736(16)31460-X
541	2.	Shaw A. The economics of African trypanosomiasis. In: Maudlin I, Holmes P, Miles
542		M, editors. The Trypanosomiases. Wallingford: CABI; 2004. pp. 369-402.
543	3.	Cecchi G, Mattioli RC. Global geospatial datasets for African trypanosomiasis
544		management: a review. Progr Against African Trypanos Tech Sci Ser. 2009;9: 1–39.
545		Available from: ftp://ftp.fao.org/docrep/fao/012/i0809e/i0809e01.pdf
546	4.	Swallow B. Impacts of trypanosomiasis on African agriculture. Progr Against African
547		Trypanos Tech Sci Ser. 2000; 1–46. Available from:
548		http://www.cabdirect.org/abstracts/20003010139.html
549	5.	Kristjanson P, Swallow B, Rowlands G, Kruska R, de Leeuw P. Measuring the costs of
550		African animal trypanosomosis, the potential benfits of control and returns to research.
551		Agric Sysytems. 1999;59: 79–98. doi:10.1016/S0308-521X(98)00086-9
552	б.	Watson RT, Zinyowera MC, Moss RH. The regional impacts of climate change: an
553		assessment of vulnerability. Intergov Panel Clim Chang. 1997; 27.
554	7.	Githeko AK, Lindsay SW, Confalonieri UE, Patz JA. Climate change and vector-borne
555		diseases: a regional analysis. Bull World Health Organ. 2000;78: 1136-1147.
556		doi:10.1590/S0042-9686200000900009

Kovats RS, Campbell-Lendrum DH, McMichel AJ, Woodward A, Cox JSH. Early

558		effects of climate change: do they include changes in vector-borne disease? Philos
559		Trans R Soc B Biol Sci. 2001;356: 1057–1068. doi:10.1098/rstb.2001.0894
560	9.	Hay SI, Cox J, Rogers DJ, Randolph SE, Stern DI, Shanks GD, et al. Climate change
561		and the resurgence of malaria in the East African highlands. Nature. 2002;415: 905-
562		909. doi:10.1038/415905a
563	10.	Zhou G, Minakawa N, Githeko AK, Yan G. Association between climate variability
564		and malaria epidemics in the East African highlands. Proc Natl Acad Sci. 2004;101:
565		2375–2380. doi:10.1073/pnas.0308714100
566	11.	Shanks G, Hay S, Omumbo J, Snow R. Malaria in Kenya's western highlands. Emerg
567		Infect Dis. 2005;11: 1425–1432.
568	12.	Wandiga SO, Opondo M, Olago D, Githeko A, Githui F, Downs T, et al. Vulnerability
569		to climate induced highland malaria in East Africa AIACC Working Paper No. 25.
570		AIACC Work Pap. 2006; 47.
571	13.	Omumbo JA, Lyon B, Waweru SM, Connor SJ, Thomson MC. Raised temperatures
572		over the Kericho tea estates: Revisiting the climate in the East African highlands
573		malaria debate. Malar J. 2011;10: 1–16. doi:10.1186/1475-2875-10-12
574	14.	Jones P, New M, Parker D, Martin S, Rigor G. Surface air temperature and its
575		variations over the last 150 years. Rev Geophys. 1999;37: 173-199.
576		doi:10.1029/1999RG900002
577	15.	Parmesan C, Yohe G. A globally coherent fingerprint of climate change impacts across
578		natural systems. Nature. 2003;421: 37-42. doi:10.1038/nature01286
579	16.	Altizer S, Ostfeld R, Johnson P, Kutz S, Harvell C. Climate change and infectious
580		diseases: from evidence to a predictive framework. Science. 2013;341: 514-519.

- doi:10.1126/science.1239401
- 582 17. Metcalf C, Walter K, Wesolowski A, Buckee C, Shevliakova E, Tatem A, et al.
 583 Identifying climate drivers of infectious disease dynamics: recent advances and
 584 challenges ahead. Proc R Soc London B Biol Sci. 2017;284.
- 585 doi:10.1098/rspb.2017.0901
- Peterson AT, Shaw J. Lutzomyia vectors for cutaneous leishmaniasis in Southern
 Brazil: ecological niche models, predicted geographic distributions, and climate
 change effects. Int J Parasitol. 2003;33: 919–931. doi:10.1016/S0020-7519(03)00094-
- 589

8

- 590 19. Ogden NH, Maarouf A, Barker IK, Bigras-Poulin M, Lindsay LR, Morshed MG, et al.
- 591 Climate change and the potential for range expansion of the Lyme disease vector

592 Ixodes scapularis in Canada. Int J Parasitol. 2006;36: 63–70.

- 593 doi:10.1016/j.ijpara.2005.08.016
- 20. Messina JP, Brady OJ, Pigott DM, Golding N, Kraemer MUG, Scott TW, et al. The
- 595 many projected futures of dengue. Nat Rev Microbiol. Nature Publishing Group;
- 596 2015;42: 1–10. doi:10.1038/nrmicro3430
- 597 21. Kula N, Haines A, Fryatt R. Reducing vulnerability to climate change in sub-Saharan
 598 Africa: the need for better evidence. PLoS Med. 2013;10.
- 599 doi:10.1371/journal.pmed.1001374
- White J. Some notes on the history and customs of the Urungwe District. South Rhod
 Nativ Aff Dep Annu. 1971;10: 33–72.
- 602 23. Hansen MC, Potapov P V, Moore R, Hancher M, Turubanova SA, Tyukavina A, et al.
- High-resolution global maps of 21st-century forest cover change. Science. 2013;342:

- 850-3. doi:10.1126/science.1244693
- 605 24. Dunham K, Mackie C, Nyaguse G. Aerial survey of elephants and other large
- herbivores in the Zambezi Valley (Zimbabwe): 2014. Gt Elephant Census, Seattle,
- 607 USA. 2015; 118.
- 608 25. Vale GA, Cumming DHM. The effects of selective elimination of hosts on a
- 609 population of tsetse flies (Glossina morsitans morsitans Westwood (Diptera,
- 610 Glossinidae)). Bull Entomol Res. 1976;66: 713–729.
- 611 doi:10.1017/S0007485300010774
- 612 26. Phelps R, Burrows P. Puparial duration in Glossina morsitans orientalis under
- 613 conditions of constant temperature. Entomol Exp Appl. 1969;12: 33–43.
- 614 27. Phelps R. The effect of temperature on fat consumption during the puparial stages of
- 615 Glossina morsitans morsitans Westw. (Dipt., Glossinidae) under laboratory conditions,
- and its implication in the field. Bull Entomol Res. 1973;62: 423.
- 617 doi:10.1017/S0007485300003953
- 618 28. Hargrove J. Reproductive rates of tsetse flies in the field in Zimbabwe. Physiol
- Entomol. 1994;19: 307–318. doi:10.1111/j.1365-3032.1994.tb01057.x
- 620 29. Hargrove J. Factors affecting density-independent survival of an island population of
- tsetse flies in Zimbabwe. Entomol Exp Appl. 2001;100: 151–164.
- 622 doi:10.1023/A:1019271727810
- 30. Hargrove J. Tsetse population dynamics. In: Maudlin I, Holmes P, Miles M, editors.
 The Trypanosomiases. CABI Publishing; 2004. pp. 113–135.
- 625 31. Pilson R. A simple apparatus for exposing tsetse flies to insecticide deposits. Rhod J
- 626 Agric Res. 1968;6: 109–112.

- Thomson M. The effect on tsetse flies (Glossina spp.) of deltamethrin applied to cattle
 either as a spray or incorporated into ear-tags. Trop Pest Manag. 1987;33: 329–335.
 doi:https://doi.org/10.1080/09670878709371177
- 630 33. Torr S, Holloway M, Vale G. Improved persistence of insecticide deposits on targets
- 631 for controlling Glossina pallidipes (Diptera: Glossinidae). Bull Entomol Res. 1992;82:
- 632 525–533. doi:https://doi.org/10.1017/S0007485300042620
- 633 34. Fenn R. Effect of physiological age and pregnancy stage on the tolerance of Glossina
- species to aerosol and topical application of endosulfan and the consequences for aerial
- 635 control. Trop Pest Manag. 1992;38: 453–458.
- 636 doi:https://doi.org/10.1080/09670879209371745
- 637 35. Mangwiro T, Torr S, Cox J, Holloway M. The efficacy of various pyrethroid
- 638 insecticides for use on odour-baited targets to control tsetse. Med Vet Entomol.
- 639 1999;13: 315–323. doi:10.1046/j.1365-2915.1999.00165.x
- 640 36. Njoroge M, Tirados I, Lindsay S, Vale G, Torr S, Fillinger U. Exploring the potential
- of using cattle for malaria vector surveillace and control: a pilot study in western
- 642 Kenya. Parasites and Vectors. 2017;10. doi:https://doi.org/10.1186/s13071-016-1957-8
- 643 37. Hansen J, Ruedy R, Glascoe J, Sato M. GISS analysis of surface temperature change. J
 644 Geophys Res Atmos. 1999;104: 30997–31022. doi:10.1029/1999JD900835
- 645 38. Hansen J, Ruedy R, Sato M. A closer look at United States and global surface
- 646 temperature change. J Geophys Res. 2001;106: 947–963. doi:10.1029/2001JD000354
- 647 39. Hansen J, Sato M, Ruedy R, Lo K, Lea D, Medina-elizade M. Global temperature
- 648 change. Proc Natl Acad Sci. 2006;103: 14288–14293. doi:10.1073/pnas.0606291103
- 649 40. Hyndman R, Khandakar Y. Automatic time series forecasting: the forecast package for

650 R. J Stat Softw. 2008;26: 1–22.

- 41. R Core Team. R: A language and environment for statistical computing. R Foundation
 for Statistical Computing; 2014.
- 42. Pilson R, Leggate B. A diurnal and seasonal study of the feeding activity of Glossina
 pallidipes Aust. Bull Entomol Res. 1962;53: 541–550 pp.
- 655 doi:10.1017/S0007485300048306
- 43. Hargrove J, Vale G. Catches of Glossina morsitans morsitans Westwood and G.
- 657 pallidipes Austen (Diptera: Glossinidae) in odour-baited traps in riverine and
- deciduous woodlands in the Zambesi Valley of Zimbabwe. Bull Entomol Res.
- 659 1980;70: 571. doi:10.1017/S0007485300007872
- 660 44. Phelps R, Burrows P. Prediction of the pupal duration of Glossina morsitans orientalis
 661 Vanderplank under field conditions. J Appl Ecol. 1969;6: 323–337.
- 45. Hargrove J, Ouifki R, Ameh J. A general model for mortality in adult tsetse (Glossina
- 663 spp.). Med Vet Entomol. 2011;25: 385–94. doi:10.1111/j.1365-2915.2011.00953.x
- 46. Amarasekare P, Sifuentes R. Elucidating the temperature response of survivorship in
 insects. Funct Ecol. 2012;26: 959–968. doi:10.1111/j.1365-2435.2012.02000.x
- 47. Phelps R, Clarke G. Seasonal elimitation of some size classes in males of Glossina
 morsitans morsitans Westw. (Diptera, Glossinidae). Bull Entomol Res. 1974;64: 313–
 324.
- 48. Hargrove J, Ackley S. Mortality estimates from ovarian age distributions of the tsetse
- 670 fly Glossina pallidipes Austen sampled in Zimbabwe suggest the need for new
- analytical approaches. Bull Entomol Res. 2015;105: 294–304.
- 672 doi:10.1017/S0007485315000073

673	49.	Ackley S, Hargrove J. A dynamic model for estimating adult female mortality from
674		ovarian dissection data for the tsetse fly Glossina pallidipes Austen sampled in
675		Zimbabwe. PLoS Negl Trop Dis. 2017;11: 1–20. doi:10.1371/journal.pntd.0005813
676	50.	Belisle J. Convergence theorems for a class of simulated annealing algorithms on R. J
677		Appl Probab. 1992;29: 885–895.
678	51.	Nelder JA, Mead R. A simplex method for function minimization. Comput J. 1965;7:
679		308–313. doi:10.1093/comjnl/7.4.308
680	52.	Hargrove J, Williams B. Optimized simulation as an aid to modelling, with an
681		application to the study of a population of tsetse flies, Glossina morsitans morsitans
682		(Diptera: Glossinidae). Bull Entomol Res. 1998;88: 425.
683		doi:10.1017/S0007485300042164
684	53.	Vale G, Hargrove J. Predicting the effect of climate change on the abundance and
685		distribution of tsetse flies. Q Updat Epidemiol from South African Cent Epidemiol
686		Model Anal (SACEMA). 2015; 1–3.
687	54.	Randolph S, Williams B, Rogers D, Connor H. Modelling the effect of feeding-related
688		mortality on the feeding strategy of tsetse (Diptera: Glossinidae). Med Vet Entomol.
689		1992;6: 231–240.
690	55.	Hargrove J, Williams B. A cost-benefit analysis of feeding in female tsetse. Med Vet
691		Entomol. 1995;9: 109–119.
692	56.	Vale G. Artificial refuges for tsetse flies (Glossina spp.). Bull Entomol Res. 1971;61:
693		331–350. doi:10.1017/S0007485300057874
694	57.	Muzari M, Hargrove J. Artificial larviposition sites for field collections of the puparia
695		of tsetse flies Glossina pallidipes and G. m. morsitans (Diptera: Glossinidae). Bull

Entomol Res. 2005;95: 221–229.

58. English S, Cowen H, Garnett E, Hargrove J. Maternal effects on offspring size in a
natural population of viviparous tsetse fly. Ecol Entomol. 2016;41: 618–626.
doi:10.1111/een.12333

- 700 59. Hargrove J, Muzari M, English S. How maternal investment varies with environmental
- factors and the age and physiological state of wild tsetse Glossina pallidipes and

Glossina morsitans morsitans. R Soc Open Sci. 2018;5. doi:10.1098/rsos.171739

- 703 60. Vale G. The responses of tsetse flies (Diptera, Glossinidae) to mobile and stationary
- baits. Bull Entomol Res. 1974;64: 545. doi:10.1017/S0007485300035860
- 705 61. World Health Organization. Mapping the distribution of human African
- 706trypanosomiasis. In: Atlas of human African trypanosomiasis [Internet]. [cited 1 Jan
- 707 2018]. Available from:
- 708 http://www.who.int/trypanosomiasis_african/country/foci_AFRO/en/
- Bursell E. Dispersal and concentration of Glossina. In: Mulligan H, editor. The African
 Trypanosomiases. London, George Allen and Unwin; 1970. pp. 382–394.
- 711 63. Odiit M, Coleman PG, Liu WC, McDermott JJ, Fevre EM, Welburn SC, et al.
- 712 Quantifying the level of under-detection of *Trypanosoma brucei rhodesiense* sleeping
- sickness cases. Trop Med Int Heal. 2005;10: 840–849.
- 64. Cunningham L, Lingley J, Haines L, Ndung'u J, Torr S, Adams E. Illuminating the
- 715 Prevalence of Trypanosoma brucei s.l. in Glossina Using LAMP as a Tool for
- 716 Xenomonitoring. PLoS Negl Trop Dis. 2016;10: 1–13.
- 717 doi:10.1371/journal.pntd.0004441
- 718 65. Lord J, Mthombothi Z, Lagat V, Atuhaire F, Hargrove J. Host-seeking efficiency can

720		to host density decline. PLoS Negl Trop Dis. 2017;11: 1–14.
721		doi:10.1371/journal.pntd.0005730
722	66.	Wittemyer G, Elsen P, Bean W, Coleman A, Burton O, Brashares J. Accelerated
723		human population growth at protected area edges. Science. 2008;321: 123-126.
724		doi:10.1126/science.1158900
725	67.	Bourn D, Reid R, Rogers D, Snow B, Wint W. Environmental change and the
726		autonomous control of tsetse and trypanosomiasis in sub-Saharan Africa.
727		Environmental Research Group Oxford Ltd; 2001.
728	68.	Gething P, Smith D, Patil A, Tatem A, Snow R, Hay S. Climate change and the global
729		malaria recession. Nature. 2010;465: 342-345. doi:10.1038/nature09098
730	69.	World Health Organisation. Number of new reported cases (T.b. gambiense) data by
731		country. In: Global Health Observatory data repository [Internet]. 2017. Available
732		from: http://apps.who.int/gho/data/node.main.A1636?lang=en
733	70.	Courtin F, Rayaissé J, Tamboura I, Serdébéogo O, Koudougou Z, Solano P, et al.
734		Updating the northern tsetse limit in Burkina Faso (1949-2009): Impact of global
735		change. Int J Environ Res Public Health. 2010;7: 1708–1719.
736		doi:10.3390/ijerph7041708
737	71.	Courtin F, Jamonneau V, Duvallet G, Garcia A, Coulibaly B, Doumenge J, et al.
738		Sleeping sickness in West Africa (1906-2006): Changes in spatial repartition and
739		lessons from the past. Trop Med Int Heal. 2008;13: 334–344. doi:10.1111/j.1365-
740		3156.2008.02007.x
741	72.	Franco J, Cecchi G, Priotto G, Paone M, Diarra A, Grout L, et al. Monitoring the

explain population dynamics of the tsetse fly Glossina morsitans morsitans in response

- elimination of human African trypanosomiasis: Update to 2014. PLoS Negl Trop Dis.
- 743 2017;11: 1–26. doi:10.1371/journal.pntd.0005585

746 Supporting information

- 747
- 748 **S1 Fig. Study site location.** Rekomitjie Research Station, located within Mana Pools
- National Park. Also showing woodland cover (2002) and loss (2000 2014) as estimated by
- 750 Hansen et al. [23]. Source: Hansen/UMD/Google/USGS/NASA.

751 S2 Fig. Adult temperature-dependent mortality.

- 752 **S1 Table. Sensitivity analysis.** Effect of varying pupal temperature-dependent mortality
- 753 parameters on fitted parameter estimates.
- 754 S1 Text. Results of alternative model fits.